Author's response to reviews

Title: Effect of foot orthoses on lower extremity kinetics during running: a systematic literature review.

Authors:

Andrew McMillan (am3mcmillan-sexton@students.latrobe.edu.au)
Craig Payne (C.Payne@latrobe.edu.au)

Version: 2 Date: 14 September 2008

Author's response to reviews:

Dear Sir / Madam,

Please find attached a revision of the manuscript with one minor amendment: four tables have been adjusted to include the headings 'Author' and 'Orthosis Design'.

This is the only amendment made.

Yours Sincerely,

Andrew McMillan